Contents lists available at ScienceDirect

Applied Energy

journal homepage: www.elsevier.com/locate/apenergy

Seeking for a climate change mitigation and adaptation nexus: Analysis of a long-term power system expansion

Kamia Handayani^{a,b,*}, Tatiana Filatova^{a,c}, Yoram Krozer^{a,d}, Pinto Anugrah^e

^a Department of Governance and Technology for Sustainability, University of Twente, Drienerlolaan 5, 7522 NB Enschede, the Netherlands

^b PT PLN (Persero), Jl. Trunojoyo Blok MI/135, Kebayoran Baru, Jakarta Selatan, Indonesia

^c School of Information, Systems and Modeling, Faculty of Engineering and IT, University of Technology Sydney, 15 Broadway, Ultimo, NSW 2007, Australia

^d Sustainable Innovations Academy, Amsterdam, the Netherlands

e Department of Electrical Engineering, Universitas Negeri Padang, Jalan Prof. Dr. Hamka, Air Tawar Padang, Sumatera Barat, Indonesia

HIGHLIGHTS

- Integrated analysis of climate change mitigation and adaptation in the power sector.
- Climate change constraints are incorporated into simulations of a power system expansion.
- The impacts of climate change on the electricity demand and supply are quantified.
- WEAP and LEAP software tools are used.
- Results indicate the nexus between climate change mitigation and adaptation.

ARTICLE INFO

Keywords: Climate change mitigation Climate change adaptation Power sector Paris Agreement LEAP Energy sector

ABSTRACT

Reductions in carbon emissions have been a focus of the power sector. However, the sector itself is vulnerable to the impacts of global warming. Extreme weather events and gradual changes in climate variables can affect the reliability, cost, and environmental impacts of the energy supply. This paper analyzed the interplay between CO_2 mitigation attempts and adaptations to climate change in the power sector using the Long-range Energy Alternative Planning System (LEAP) model. This paper presented a novel methodology to integrate both CO_2 mitigation goals and the impacts of climate change into simulations of a power system expansion. The impacts on electricity supply and demand were quantified, based on historical climate-related impacts revealed during fieldwork and existing literature. The quantified effects, together with climate mitigation targets, were then integrated into the LEAP modeling architecture. The results showed a substantial alteration in technology composition and an increase in installed capacities driven by the joint climate mitigation–adaptation efforts when compared with the scenario without mitigation-adaptation (reference). Furthermore, an increase in CO_2 emissions was observed under the mitigation-adaptation scenario compared with the mitigation only scenario, indicating that the power sector's adaptations for climate change are likely to hinder CO_2 mitigation efforts. Therefore, a nexus between mitigation and adaptation should be exploited in the policy development for a low-carbon and climate-resilient power system.

1. Introduction

The energy sector contributes significantly to global warming. Accordingly, most of the Nationally Determined Contributions (NDCs) submitted by Parties to the United Nations Framework Convention on Climate Change (UNFCCC) included energy sector emission reduction components, as strategies for meeting their pledges under the Paris Agreement [1]. However, the energy sector is also vulnerable to the impacts of climate change (CC). Extreme weather events and gradual changes in climate variables can affect the reliability, cost, and environmental impacts of the energy supply [2,3]. Therefore, meeting climate mitigation goals while simultaneously coping with CC impacts represents a tremendous challenge for the energy sector. In developing countries, this challenge often coincides with vital electrification objectives, caused by rapid growth in electricity demand.

The societal challenge of transitioning from fossil fuels to renewable

https://doi.org/10.1016/j.apenergy.2019.114485

Received 12 July 2019; Received in revised form 12 December 2019; Accepted 29 December 2019

 $0306-2619/ \odot 2020$ The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/BY-NC-ND/4.0/).





Check fo

^{*} Corresponding author at: Room RA 1161, Ravelijn Building, P.O. Box 217, 7500 AE Enschede, the Netherlands. *E-mail address:* k.handayani@utwente.nl (K. Handayani).

Nomenclature		T_{sw}	seawater temperature
		u_2	wind speed
Δ	slope vapor pressure curve		
β	the power temperature coefficient	Abbreviation	
η_T	the temperature derating factor		
γ	psychrometric constant	EURO-CORDEX	European Coordinated Regional Downscaling
$C_p(T_{sw})$	condenser pressure at seawater temperatureT _{sw}		Experiment
CF_{Tm}	annual capacity factor at temperature T_m	GEP	Generation Expansion Planning
$Dp(T_m)$	annual peak demand at temperature T_m	HiREPS	High-Resolution Power System
ea	actual vapor pressure	LEAP	Long-range Energy Alternative Planning system
e_s	saturation vapor pressure	MAED	Model for Analysis of Energy Demand
E_{Ref}	annual energy generation at referent temperature	MARKAL	MARket ALlocation
ET	evapotranspiration	MESSAGE	Model for Energy Supply Strategy Alternatives and
E_{Tm}	annual energy output		their General Environmental Impact
G	soil heat flux density	POLES	Prospective Outlook on Long-term Energy Systems
Р	installed capacity	TIAM-ECN	TIMES Integrated Assessment Model of the Energy
P(T)	power output at temperatureT		research Centre of the Netherlands
R_n	net radiation at the crop surface	TIMES	The Integrated MARKAL/EFOM System
T_m	the mean air temperature	WEAP	Water Evaluation And Planning system
T_{Ref}	the referent temperature		

energy in response to the Paris Agreement triggered significant research efforts. Grande-Acosta and Islas-Samperio [4] provided alternative scenarios for decarbonizing the Mexican power sector, by assessing 36 mitigation options, using the Long-range Energy Alternative Planning System (LEAP). Handayani et al. [5] also employed LEAP to analyze scenarios that would allow the Indonesian power system to meet both electrification and CC mitigation goals. Dalla Longa and van der Zwaan [6] adopted the TIAM-ECN model to analyze the roles played by low carbon technologies in attempts to achieve the Paris Agreement climate targets in Kenya. A study of the Malaysian power sector, conducted by Haiges et al. [7], utilized the TIMES model to assess long-term power generation options while also considering CC mitigation. Guo et al. [8] analyzed the de-carbonization of the Chinese power sector using a multi-region dispatch model, revealing unintended consequences of the de-carbonization pathway, such as the disturbance in the stability and integrity due to intermittent renewable energy generation. Overall, these studies have indicated that the Paris Agreement targets are achievable but are also associated with various consequences, such as increasing costs and alterations in technology and energy mixes.

While mitigation efforts are vital, the adverse impacts of CC already affect various sectors of social-economic systems that may either facilitate or hinder the mitigation efforts made by the power sector [3]. Previous studies have paid little attention to adverse CC impacts that are expected to impact entire elements of the power system. Electricity supply and demand are the first factors likely to be affected. On the supply side, CC impacts include changes in water availability and the seasonality of hydropower, alterations in wind speed frequencies and distributions, reductions in solar cell efficiency, declines in generation cycle efficiencies and the availability of cooling water for thermal power plants, and the failures and capacity reductions of transmission and distribution lines. On the demand side, CC alters heating and cooling patterns [2,3,9]. Neglecting these impacts can undermine efforts to decarbonize the energy sector [3].

Furthermore, the Fifth Assessment Report (AR5) of the Intergovernmental Panel on Climate Change (IPCC) has recognized that the CC impacts and adaptation responses of the energy system represent research gaps and urge the integration of these issues into assessments of climate stabilization pathways [10]. Because CC impacts could influence the effectiveness of mitigation options, the integration of these factors crucial when developing mitigation plans. AR5 also highlighted the scarcity of publications regarding CC impacts, adaptations, and vulnerability in developing countries [11]. This paper aimed to address these gaps in the literature by conducting a holistic analysis of lowcarbon pathways for the power sector while accounting for adverse CC impacts. Both CC mitigation goals and CC impacts were integrated into modeling simulations of power system expansions, using the primary power system of Indonesia, Java-Bali, as a case study. Indonesia is an emerging and developing economy, with fast-rising electricity demand, and aims to develop a low-carbon energy system, targeting a 23% share of renewable energy in the national energy mix, by 2030, and a 31% by 2050. These targets are in line with the country's NDC, which pledge to reduce CO_2 emissions by 29%.

LEAP was used to conduct long-term expansion simulations for the Java-Bali power system. The modeling tool was chosen over 30 models potentially suitable for the analysis of this paper, owing to its ability to accommodate the technological complexity of the power system, its support for alternative scenario projections, its least-cost optimization modeling of power system expansions, and its ability to calculate CO_2 emissions from various power generation technologies. Furthermore, LEAP is freely accessible to students and academia in developing countries and has user-friendly features. The models' screening procedure and the validation of the Indonesian LEAP model were reported in our previous work [5].

Prior to commencing the simulations, the impacts of CC on electricity demand and supply were quantified based on intensive fieldwork performed in Indonesia - which revealed the past impacts of climaterelated events - and secondary data from the literature. The quantified effects, together with the renewable energy targets, were then converted into constraints on the technological characteristics of power plants, within the LEAP optimization modeling architecture.

The paper offers two innovative contributions to the academic literature and real-world practices. First, to the best of our knowledge, this research is the first attempt to jointly consider CC mitigation goals and CC impacts on various types of power generation technologies and electricity demand, while modeling power system expansions using LEAP. An explicit link between the sectoral efforts to pursue a low-carbon pathway and its adaption to adverse CC impacts requires novel methodological development. This article makes steps in filling this important gap in the literature by outlining the methodology and illustrating its feasibility in a case study. Second, this paper adds to the scarce literature regarding the impacts of CC, particularly the vulnerability and adaptation of the power sector in developing countries, by quantifying sector-wise costs of CC. As such, it may serve as a reference for the development of policies that facilitate the formation of a low-carbon and climate-resilient power sector.

The remaining portions of this paper proceed as follows. Section 2

describes the existing literature regarding power system expansion modeling under CC conditions and outlines the existing knowledge gaps. Section 3 describes the employed methodology, and Section 4 describes the LEAP modeling results for each scenario. Section 5 discusses the significance of these findings. Finally, Section 6 provides conclusions, discusses the limitations of this study, and makes recommendations for future work.

2. Modeling climate change impacts on the power sector

Increasingly, the idea that the electricity sector is vulnerable to the effects of CC being recognized, as is the need to quantify these impacts using models. Tobin et al. [12] investigated the impacts of CC on four electricity supply technologies (wind, solar photovoltaic [PV], hydro, and thermal power) in 28 European Union (EU) countries, based on the European Coordinated Regional Downscaling Experiment (EURO-CORDEX) regional climate model projections. Other studies have focused primarily on the impacts of CC for thermoelectric power plants. Kern and Characklis [13] used a systemic framework to assess the impacts of CC-associated drought on the financial exposure of a major utility in the Southeastern part of the United States (U.S.). Cook et al. [14] developed a methodology for predicting the risk that power plants will violate thermal pollution limits, due to climate-induced drought and heatwaves in the Upper Mississippi River Basin and Texas. Furthermore, van Vliet et al. [15] analyzed the impacts of CC-associated changes in temperature and water availability on thermoelectric power plants in the U.S. and Europe, using a physically-based hydrological and water temperature modeling framework, in combination with an electricity production model. Likewise, Zheng et al. [16] investigated the vulnerability of Chinese thermoelectric power plants to water scarcity problems, by performing a high-resolution evaluation and projection of spatial vulnerability. Collectively, these studies agreed that the adverse effects associated with changes in climate variables, such as temperature, and precipitation, would likely reduce the power generation capacities of thermoelectric power plants.

Adverse impacts associated with gradual changes in climate variables on wind and solar PV have been found to be limited [12,17,18]. Moreover, Lucena et al. [18] and Pašičko et al. [19] estimated increased wind power production under future climate conditions in some parts of Brazil and Croatia, respectively. However, off-shore wind power plants may require investments for adaptation measures that counteract rises in the sea level [20].

The results regarding CC impacts on hydropower have been mixed. Anugrah et al. [21] evaluated the Bayang micro hydropower system in Indonesia, using the Water Evaluation And Planning System (WEAP)-LEAP models, and found that CC will reduce power production. Guerra et al. [22] employed a two-stage stochastic approach to power system design and planning to analyze the impacts of CC on the hydro-dominant Columbian power system, which suggested that CC will reduce the capacity factor of hydropower plants in Columbia. Meanwhile, Boehlert et al. [23] evaluated the impacts of CC on hydropower generation in the U.S., using a set of linked models, which suggested a possible increase in future hydropower generation due to increased river runoff, based on high-emission scenarios in the Pacific Northwest.

Compared with other energy sources, the potential CC impacts on hydropower have been extensively incorporated into simulations of long-term power system expansion. This high level of attention is likely due to the hydropower dominance of the energy sector in some countries, the long asset lifetime associated with hydropower, and the apparent dependence of hydropower on climate-related factors [3]. Integrating potential CC impacts into simulations of power system expansions allows the associated impacts of CC to be analyzed for the entire power system. The studies that have examined CC impacts include Lucena et al. [24], for the Brazilian power sector, Teotónio et al. [25], for the Portuguese power system, Arango-Aramburo et al. [26], for the Colombian power sector, and Spalding-Fecher [27], for the Southern African Power Pool. Overall, these studies have indicated a decline in hydropower generation under CC conditions, although differences exist in the degree of decline, depending on the country of interest. Consequently, extra capacity from other energy sources may be required to compensate for a future lack of hydropower reliability.

On the demand side, previous studies have often indicated increases in electricity consumption capacity associated with CC, such as in the U.S. [28,29,30], the Southeastern Mediterranean region [31], Brazil [24], and Hong Kong [32]. A net increase in demand has also been indicated in Austria and Germany [33]. In contrast, the total energy demand of the EU is expected to decrease [34]. Similarly, a study examining Norway [35] also suggested a net decrease in energy demand.

Several studies have integrated multiple CC-associated impacts on energy demand and various types of energy production technologies into simulations of long-term power system expansions, allowing their combined effects on the system to be examined. Lucena et al. [24] incorporated expected CC-associated impacts on electricity demand, hydropower capacity, and natural gas turbine capacity into simulations of the long-term expansion of the Brazilian power sector, using Model for Analysis of Energy Demand (MAED)-Model for Energy Supply Strategy Alternatives and their General Environmental Impact (MESSAGE) models. Seljom et al. [35] employed market allocation (MARKAL) modeling to integrate CC-associated impacts on hydro and wind power, as well as energy demand, into an analysis examining the future of the Norwegian energy system. Meanwhile, Dowling [34] analyzed the European energy system, considering CC-associated impacts on energy demand and various types of power production methods, including thermoelectric, wind, solar PV, and hydroelectric power generation, by utilizing the Prospective Outlook on Long-term Energy Systems (POLES) model. A similar study examining Europe was conducted by Mima and Criqui [36], with a smaller scope, including only energy demand, thermal power plants, and hydropower. Totschnig et al. [33] employed the High-Resolution Power System (HiREPS) to integrate the impacts of CC and fuel price changes on energy demand and various types of energy production methods into simulations of future Austrian and German power sectors. Finally, Li et al. [37] incorporated the effects of projected changes in temperature, precipitation, and extreme events on electricity demand and various power supply types into simulations of power system expansion for several U.S. regions, using a robust electric power Generation Expansion Planning (GEP) optimization model. These studies have all agreed that the reliability of thermal power plants will decrease under CC conditions. Meanwhile, the CC impacts on solar PV and wind have been estimated to be minor [35,34]. The results for Norway have revealed lower estimated system costs due to a projected net decrease in demand. Furthermore, Norway is likely to benefit from an increase in hydropower production. In contrast, Brazil is likely to suffer from reduced hydropower reliability, requiring extra installed capacity, and, consequently, requiring extra investment costs.

However, to the best of our knowledge, research explicitly linking the impacts of CC with the energy sector's climate mitigation goals remains lacking. Hence, how CC impacts may interfere with the power sector's efforts to achieve national long-term climate mitigation goals remains unexplored. An integrated assessment of CC mitigation and adaptation in the power sector is especially vital for developing countries, which already face adverse CC impacts and rising electrification demands, on top of CC mitigation goals. This paper addressed this important gap by pursuing a comprehensive assessment of the nexus between electrification, CC mitigation, and CC adaptation in developing countries, using Indonesia as an example.

3. Methodology and data

This study simulated the future development of the Indonesian power system, taking into consideration both CC mitigation and adaptation. This study focused particularly on the extensive Java-Bali power system, which provides 75% of total Indonesian electricity consumption [38], serves 59% of the Indonesian population [39], and mirrors the national power sector in terms of the historical energy mix, supply, and demand [40].

To account for CC mitigation, a long-term scenario for Java-Bali power system expansion was developed that included the country's low-carbon development policy (Fig. 1, left side of the flow chart). Subsequently, LEAP, an energy system model developed by the Stockholm Environment Institute (SEI), was used to analyze the low-carbon pathway. LEAP is a prominent model that has been used in 190 countries [41], 85 UNFCCC country reports [42], and more than 70 peerreviewed journal papers [43]. The popularity of LEAP makes results using this model comparable across countries, which is especially relevant when major international agreements, such as the Paris accord, are considered. As a bottom-up energy model, LEAP offers the capability to analyze both power system expansion and climate policy scenarios, taking into account the detailed characteristics of electric power technologies. A detailed description of LEAP is provided in Appendix A.

For the adaptation¹ component, a 3-step approach was applied (Fig. 1, right side of the flow chart). First, data were collected regarding the historical impacts of severe weather events and gradual changes in climate variables on the power sector, through semi-structured interviews and focus group discussions at 10 major power plants, between February and March 2018. The fieldwork revealed that weather and climate affect all segments of the power system, which include generation, transmission, and distribution. Greater details regarding the findings from the fieldwork were presented in Handayani et al. [44]. Second, a literature review examining CC scenarios was performed to identify likely trends in temperature, precipitation, and sea surface temperatures for the Indonesian archipelago. Finally, the variables and functions necessary for the LEAP model architecture were identified to parameterize the projected impacts of CC on the Java-Bali power system.

The impacts of CC were considered in terms of both demand and supply. To consider the impacts of CC on energy demand, this paper estimated how gradual changes in temperature might affect the electricity requirements of the Java-Bali system. Although the fieldwork revealed that extreme weather events, such as heatwaves, have caused the temporary shutdown of several power plants [44], this paper excluded extreme weather events from the simulation model, due to methodological and data-associated limitations; therefore, the inclusion of these extreme weather events will be an important direction for future work.

To consider the impacts of CC on energy supply, the impacts of CC on power generation were quantified, focusing on four power plant types: coal-fired power plants (CFPPs), natural gas power plants (NGPPs), hydroelectric power plants (HEPPs), and solar photovoltaic (solar PV) power plants. CFPPs, NGPPs, and HEPPs were chosen because they constitute most of Indonesia's power generation capacity, at 50%, 28%, and 9%, respectively. Solar PV was included because this type of power generation is expected to play a significant role in Indonesia's transition to a low-carbon energy system [40]. The impacts of CC on wind power have been discussed in the literature. For example, Schaeffer et al. (2012) indicated that alterations in wind speed may influence the ability to optimize a wind energy source with a wind turbine power curve [2]. Further modeling studies may quantify these impacts and integrate them into LEAP simulations for power system expansions, enabling a more comprehensive analysis of the impacts of CC on the power system.

This paper focused on the analysis of how gradual changes in climate variables, including surface air temperatures, precipitation levels,



Fig. 1. The conceptual framework for integrating climate change mitigation and adaptation, using WEAP-LEAP models.

and sea surface temperatures, impact the power system. WEAP was employed to analyze how changes in precipitation and temperature may impact water availability for hydroelectric power plants. In addition, this present paper relied on the findings from fieldwork and from the current literature to parameterize the expected CC impacts on thermal power plants (TPPs) and solar PV.

Finally, LEAP was used to simulate the Java-Bali power system expansion, while considering the national low-carbon policy targets and the CC impacts on the power system. These simulations quantified the electricity mix and corresponding costs of climate-resilient energy systems while ensuring that the electrification, CC mitigation, and adaptation goals are being met.

3.1. Scenario development

3.1.1. Climate change scenarios

The CC projections in this study are based on the Representative Concentration Pathway (RCP) 4.5 and RCP8.5 scenarios that were adopted in the IPCC AR5. RCP4.5 is a medium-stabilization scenario that assumes a stable radiation intensity, at approximately 4.5 W/m^2 or equivalent to 650 ppm CO₂ eq., after 2100. RCP8.5 a very-high-baseline emission scenario, which assumes a constant increase in pathways leading to 8.5 W/m² of radiation intensity (more than 1,370 ppm CO2 eq.), in 2100 [45]. The CC projections for Indonesia in this study included these scenarios. The projections for temperature and precipitation (Table B.1. Appendix B) were based on the ensemble values from 35 Coupled Model Intercomparison Project (CMIP) models, which are available at the World Bank Climate Knowledge Portal website [46] and were derived from those used by Harris et al. [47]. These data are available for two periods within our study's time horizon: 2020-2039 and 2040-2059. Meanwhile, the projected sea surface temperature data were obtained from the output of a global climate model from the Institut Pierre Simon Laplace, which was retrieved from the International Pacific Research Center website² [48]. Based on these data, the effects of climate-related variables on electricity demand and the key technical characteristics of various power plants were quantified.

¹ Adaptation is defined here as variations in the power system configuration (e.g., installed capacity, technology composition), due to the CC-induced reductions in power supply reliability and changes in electricity demands.

² The Asia-Pacific Data Research Center is a part of the International Pacific Research Center at the University of Hawai'i at Mānoa, funded in part by the National Oceanic and Atmospheric Administration (NOAA).

3.1.2. Power system expansion scenarios

The three following scenarios for the expansion of the Java-Bali power system were developed for the period from 2018 to 2050.

Reference scenario (REF): a power system expansion scenario, without CC considerations. The objective of this scenario is to satisfy the growing future demand for electricity in the Java-Bali islands at the lowest overall cost.

CC mitigation scenario (CCM): this power system expansion scenario assumes a shift to a low carbon pathway, based on Indonesia's New and Renewable Energy (NRE) targets, increasing the share of renewable energy to 23%, by 2025, and 31%, by 2050. In addition to these mitigation targets, this scenario includes electrification targets but neglects adaptations to CC.

CC mitigation and adaptation scenario (CCMA): this power system expansion scenario assumes a shift to a low-carbon pathway and integrates the effects of projected CC on both electricity supply and demand. Based on the CC scenarios discussed in Section 3.1.1, two CCMA scenarios were examined:

a. CCMA RCP4.5: CC impacts on power plants were quantified based on the IPCC RCP4.5 scenario

 b. CCMA RCP8.5: CC impacts on power plants were quantified based on the IPCC RCP8.5 scenario

The Java-Bali LEAP model is described in detail in Handayani et al. [40]. In addition to the methodological advances associated with the inclusion of the CCMA scenarios, this present study updates the previous study by including the most recently available historical data regarding energy demand and supply. Accordingly, the LEAP base year has been updated from 2015 to 2017. Furthermore, the electricity demand projection was updated based on the most recently available electricity supply business plan (RUPTL 2018–2027) [49]. Likewise, the capital costs of electric power technologies have now been assumed to decrease every five years, based on the percentage of technology cost reductions reported by the World Energy Outlook [50]. Table 1 and Table 2 list the primary assumptions made for the Java-Bali LEAP model. Meanwhile, the potential renewable energy sources that were exploited by this modeling simulation are presented in Table C.1. Appendix C.

3.2. Integrating adverse CC impacts into the LEAP model

3.2.1. Demand-side

3.2.1.1. Projecting the impacts of higher temperatures on electricity demand. Prior to estimating future climate-driven electricity demands, the actual hourly load data and the peak electricity demand data were collected from the Java-Bali Load Control Center. Comparing these data to temperature data collected from 23 weather stations, retrieved from the Agency for Meteorology, Climatology, and Geophysics [61], demonstrated a significant correlation between temperature and electricity demand [44]. A first-order estimate was

performed to determine the effects of projected higher temperatures on annual peak demand, using Eq. (1).

$$Dp(T_m) = 521.55T_m + 5788.3 \tag{1}$$

where Dp(T) is the annual peak demand (in MW) and T_m is the mean air temperature (°C).

The results showed that the annual demand for electricity is likely to increase by 1.2% and 1.4% by 2020, under the CCMA RCP4.5 and CCMA RCP8.5 scenarios, respectively, compared with the REF scenario. Furthermore, by 2040, electricity demand is likely to increase by 2% and 2.8%, respectively.

3.2.2. Supply-side

As a bottom-up energy model, LEAP allows the detailed characteristics of energy technologies to be determined as input data; therefore, the technical characteristics of a power plant can be altered to integrate the effects of CC. Based on data gathered through fieldwork and from the literature that has reported the impacts of weather and climate on power plants' performances, these effects could be quantified. The most detrimental effects, such as changes in surface air temperature and precipitation, were identified and transformed into changes in the technical characteristics of power plants. The altered technical characteristics included capacity factors and efficiency. These changes were expected to influence the technology mix and installed capacities and to reflect the costs of the power system adapting to CC.

3.2.2.1. Projecting the impacts of temperature and precipitation changes on *HEPPs*. In this study, WEAP was employed to simulate future hydrology in the Citarum river basin, based on projected changes in air temperatures and precipitation. The Citarum river provides water for the two largest HEPPs in Indonesia, Saguling and Cirata, with installed capacities of 1008 MW and 797 MW, respectively (see Fig. D.1., Appendix D). WEAP is commonly used to simulate water demand, supply, flow, storage, discharge, and pollution scenarios. Therefore, this method is suitable for analyzing water availability for HEPPS under future CC scenarios. Furthermore, WEAP has a built-in link to LEAP modeling, where the hydropower availability modeled by WEAP becomes an input for LEAP. Hence, the use of both software tools together enables the dynamic analyses of CC implications for hydropower production [62].

Temperature can affect water availability in a watershed system. Higher temperatures result in larger amounts of surface water being returned to the atmosphere, through evaporation and transpiration, also known as evapotranspiration. In WEAP, evapotranspiration is calculated using the following U.S. Food and Agricultural Organization (FAO) Penman-Monteith equation, as outlined in the FAO Irrigation and Drainage Paper No. 56 [63].

$$ET = \frac{0, 408\Delta(R_n - G) + \gamma \frac{900}{T + 273} u_2(e_s - e_a)}{\Delta + \gamma(1 + 0, 34u_2)}$$
(2)

where ET is the evapotranspiration rate (mm/day), R_n is the net

Table 1		
Summary of t	he model in	put parameters.

Input Data	Value	Source
Annual demand growth 2016–2030	4–6%	Refers to the RUPTL and IEO estimates [51,49]
Transmission & distribution losses	7.9–8.5%	Refers to the draft RUKN estimates [52]
Reserve margin [*]	35%	Refers to the RUKN criteria [52]
Environmental parameter	Per technology	The IPCC Tier 1 default emission factors, embedded in the LEAP's technology database [41]
Discount rate	12%	The discount rate used by PLN [53]

Notes: IEO = Indonesia Energy Outlook, RUKN = Rencana Umum Ketenagalistrikan Nasional (National Electricity General Plan), PLN = Perusahaan Listrik Negara (State Electricity Company)

* Reserve margin is the percentage of reserve capacity relative to the capacity needed to meet the standard peak demand.

Table 2

Characteristics of the energy technologies included in the Java-Bali LEAP model.

Technology	Lifetime of power plant (years) ^a	Efficiency* (%) ^a	Maximum availability ^{**} (%) ^b	Capacity credit (%) ^{***}	Capital cost (2015 US \$/kW) ^a	Fixed OM ^{****} cost (2015 US \$/kW) ^a	Variable OM cost (2015 US \$/MWh) ^a	Fuel cost ^c (2015 US\$)
Ultra-supercritical coal	30	40	80	100	1400	31.3	2	51.8 US\$/ton
Natural gas combined cycle	25	55	80	100	800	19.2	1	7.6 US \$/MMBTU
Natural gas open cycle	20	36	80	100	700	18	1	7.6 US \$/MMBTU
Hydro	50	100	41	51	2000	6.6	1	-
Mini hydro	25	100	46	58	2400	6.6	1	-
Hydro-pumped storage	50	76	20	25	800	6.6	1	-
Geothermal	25	10 ^d	80	100	3500	30	1	-
Solar PV	20	100	17	22	2069 ^e	24.8 ^e	0.4 ^b	-
Wind power	20	100	28	35	2200	44 ^d	0.8 ^b	-
Nuclear	40 ^f	34	85 ^g	100	6000	164 ^d	8.6 ^g	9.33 US\$/MWh ^f
Biomass	20 ^h	35 ^d	80	100	2228 ^d	78 ^d	6.5 ^b	11.67 US\$/ton ^h

* Efficiency is defined as the percentage ratio of energy outputs to feedstock energy inputs in each process.

** Maximum availability in LEAP is defined as the ratio of the maximum energy produced to the potential production if the process ran at full capacity for a given period (expressed as a percentage). The 'maximum availability' data, together with the installed capacity data for a power plant, determine the annual electricity production of the power plant [41].

*** Capacity credit in LEAP is defined as the fraction of the rated capacity considered firm for calculating the reserve margin. The values are calculated based on the ratio of availability for the intermittent plant to the availability for a standard thermal plant [41].

**** OM: Operation and Maintenance. Sources:

^a PLN[54].

^b DEN [55].

^c PLN [56].

^d OECD/IEA [50].

^e ACE [57].

^f Rothwell & Rust [58].

^g IEA & NEA [59].

^h IRENA [60].

radiation at the crop surface level $(MJ/m^2/day)$, *G* is the soil heat flux density $(MJ/m^2/day)$, *T* is the mean air temperature (°C), u_2 is the wind speed (m/s), e_s is the saturation vapor pressure (kPa), e_a is the actual vapor pressure (kPa), Δ is the slope vapor pressure curve (kPa/°C), and γ is the psychrometric constant (kPa/°C). Detailed descriptions of the WEAP methodology are provided in Sieber [64] and Yates et al. [65]. The WEAP input data used for the HEPPs simulations in this study can be found in Table D.1, Appendix D.

The results showed that under the CCMA RCP4.5 scenario, the average annual capacity factors for both the Saguling and Cirata HEPPs varied across years when compared with the REF scenario (Fig. D.2., Appendix D). Under the CCMA RCP4.5 scenario, both Saguling and Cirata were projected to produce slightly less electricity (<1%) than under the REF scenario from 2022 to 2039. In contrast, their electricity production increases relative to the REF scenario starting in 2040, resulting in the aggregate HEPP production over the study period being higher under the CCMA RCP4.5 scenario than under the REF scenario. No significant difference in the annual average capacity factor over time was observed between the REF and RCP8.5 scenarios. In addition, the aggregate HEPP production for the RCP8.5 scenario was slightly lower than that for the REF scenario. These results were extrapolated to the remaining HEPPs when simulating the Java-Bali power system expansion using the LEAP-WEAP combination.

3.2.2.2. Sea surface water temperatures and the efficiencies of CFPPs. A cooling water system is an essential component of CFPPs. CFPPs in the Java-Bali power system rely on seawater for their cooling systems and are likely to be impacted by the higher seawater temperatures expected under CC conditions. The CFPPs in Indonesia employ once-through cooling systems, which circulate seawater through pipes to absorb heat from steam in condensers. The warmer water is then discharged back to the sea. A higher cooling water temperature results in increased condenser pressures, which reduces the energy efficiency of the condenser [66]. Based on data from the Paiton PJB CFPP, which is a

typical CFPP that uses a once-through cooling system, the relationship described by Eq. (3), between cooling water temperature and the performance of a condenser vacuum, was identified [44].

$$C_p(T_{sw}) = -3.4339T_{sw} + 796.33 \tag{3}$$

where $C_p(T_{sw})$ is the condenser pressure (mmHg) and T_{sw} is the seawater temperature (°C).

Furthermore, this relationship was combined with data regarding the primary energy consumption of the power plant, revealing that for every 1 °C increase in the cooling water temperature above 25 °C, the power plant efficiency drops by 0.32%. This relationship was used to estimate changes in CFPP efficiencies following the anticipated CC-associated changes in the surface seawater temperatures. The results showed that reductions in the efficiencies of Java-Bali CFPPs are expected to range from 0.1% to 0.4%, under the RCP4.5 scenario, and from 0.03% to 0.3%, under the RCP8.5 scenario. These drops in efficiency are expected to increase fuel consumption, leading to increased fuel costs and CO_2 emissions.

3.2.2.3. Precipitation and capacity factors for CFPPs. Our fieldwork revealed that heavy precipitation dampens coal stored in open storage areas, reducing the burning efficiency of CFPPs [44]. Damp coal is a particular issue during the wet season when precipitation is intense and frequent. Suralaya CFPP, the largest CFPP in Indonesia, provided operation disruption data from 2011 to 2017, which included past disruptions caused by wet coals. Based on these data, average annual reductions in CFPP capacities were calculated, based on reductions in energy generation that were attributable to wet coals.

Furthermore, historical data regarding CFPP disruptions and local precipitation were combined with projected precipitation estimates under CC conditions to estimate the expected reductions in CFPP capacities due to future precipitation events. The results indicated that the CFPP capacity factor will be reduced by 0.29% and 0.30%, respectively, by 2020 and 2040. These estimates were adopted in the LEAP

simulations for the remaining Java- Bali CFPPs.

3.2.2.4. Ambient air temperatures and power outputs of NGPPs. Two types of NGPP technologies are utilized in the Java-Bali power system: open cycle gas turbine (OCGT), which produces electricity solely from a gas combustion turbine, and combined cycle gas turbine (CCGT), which produces electricity from the gas combustion turbine and generates extra electricity by routing waste heat from the gas turbine to a nearby steam turbine. High ambient air temperatures primarily affect the operation of NGPP gas turbines because gas turbine operation intakes ambient air for the compressor and routes the air into the burning chamber. Higher ambient air temperatures reduce the air density, which, in turn, reduces the burning efficiency and power outputs [67]. According to the field data collected from Muara Karang NGPP, every 1 °C increase in ambient air temperature above 16 °C reduces the power output of the OCGT by 0.6%. Accordingly, the power output of the Muara Karang OCGT correlates with air ambient temperatures, as described by Eq. (4). These data are comparable with previously reported findings in the literature, which indicated that each 1 °C increase in temperature above 30 °C reduced gas turbine power outputs by 0.50–1.02% [68,69,70]. Furthermore, the literature indicated that the overall net reduction in CCGT power outputs ranges from 0.3% to 0.6%. Here, the power outputs of CCGTs were assumed to drop by 0.45%, which is the middle value of the range. Accordingly, the power output of CCGTs correlated with ambient air temperatures as described by Eq. (5).

Furthermore, the monthly energy outputs of OCGTs and CCGTs were estimated under CC scenarios based on average monthly temperature projections, using Eq. (4) and Eq. (5), respectively. The annual capacities for NGPPs under CC scenarios were calculated using Eq. (6). The results showed that the expected reductions in the capacities of the Java-Bali NGPPs by 2040 ranged from 0.8% to 4.3%.

$$P(T) = P * (-0.006T + 1.0933) \tag{4}$$

$$P(T) = P * (-0.0043T + 1.0662)$$
(5)

$$CF_{Tm} = \frac{E_{Tm}}{Plantcapacity(MW) * 8760hours}$$
(6)

where P(T) is the power output, at temperature T (MW), P is the installed capacity (MW), E_{Tm} is the annual energy output (MWh), T_m is the mean air temperature (°C), and CF_{Tm} is the capacity factor.

3.2.2.5. Air temperatures and solar PV efficiencies. Increased air temperature may negatively affect the performance of solar PVs, manifested as a decrease in the PV efficiency, determined by the temperature coefficient, which is a correction factor for efficiency as a function of temperature [71]. The estimate reported by Pašičko et al. [19] was utilized, in which every 1 °C increase in air temperature decreases the PV efficiency by 0.5%, relative to the referent value at 25 °C. This value is consistent with the average temperature coefficient specified by Dubey et al. [72]. Based on this assumption, the temperature derating factor of the Java-Bali solar PV caused by CC-associated temperature increases can be estimated using Eq. (7)³, and subsequent reductions in the annual power generation by solar PV can be calculated using Eq. (8).

$$\eta_T = 1 - [\beta (T_m - T_{Ref})]$$
(7)

$$E_{T_m} = \eta_T x E_{Ref} \tag{8}$$

where η_T is the temperature derating factor, β is the power temperature coefficient, Tm is the mean air temperature (°C), T_{Ref} is the referent temperature, E_{Tm} is the annual energy generation (MWh) and E_{Ref} is the

annual energy generation at referent temperature (MWh).

The results indicated that the derating of solar PV outputs under future expected air temperatures will result in solar PV capacity being reduced by 0.5% to 0.6%, during the 2020 s, and by 0.8% to 1.1%, during the 2040 s.

4. Climate change mitigation-adaptation synergies: LEAP results

The following sections discuss the outputs of the CC mitigationadaptation integrated framework, using the WEAP and LEAP models when applied to the Java-Bali power system. First, the added capacity and electricity mix for each scenario are presented, followed by discussions regarding their associated costs and CO_2 emissions.

4.1. Installed capacity and electricity mix

4.1.1. Reference scenario

Under the REF scenario, which assumes the least-cost electrification pathway, without considering CC mitigation and adaptation strategies, the capacity of fossil fuel technologies increases rapidly. The coal and natural gas capacities reach 115 GW and 53 GW, by 2050, respectively, which represent six-fold and five-fold increases, respectively, compared with their capacities in 2017. As a result, the capacity mix during the study period becomes heavily fossil fuel-dependent (Fig. 2a). In 2050, coal maintains its dominance under the REF scenario (Fig. 2b), comprising 72.4% of the Java-Bali electricity mix, followed by natural gas (20.7%), hydro (3.8%), and geothermal (3%) production.

4.1.2. Climate change mitigation scenario

The results showed that the implementation of the NRE policy (CC mitigation targets, as assumed by the CCM scenario) is likely to dramatically alter the Java-Bali capacity mix. In 2050, the coal capacity under the CCM scenario is expected to reduce by 41% compared with the REF scenario. These reductions in coal capacity are likely to be compensated by increases in the capacities of natural gas (23%), hydro (7%), and geothermal (27%) as well as by the penetration of solar (117.7 GW), biomass (7.4 GW), and wind (1.6 GW). With this capacity mix, all of the hydro, geothermal, and biomass potentials in Java-Bali islands (Table C.1, Appendix C) will be utilized, leaving wind and solar as the only potentially available renewable energy sources.

Accordingly, under the CCM scenario, the Java-Bali electricity mix is expected to mimic the changes in its capacity mix. Although fossil fuels supply most of the demand for electricity in the base year, their share is gradually reduced over time. Complying with the NEP-RE target, by 2025, renewables compose 23% of the electricity mix, shared among geothermal (9.6%), hydro (5.6%), biomass (6.5%), wind (1%), and solar (0.3%) (Fig. 3), Furthermore, by 2050, the renewable share increases to 31%, as targeted by the NRE policy. Interestingly, the electricity production from solar is expected to encompass 17.4% of the total electricity mix in 2050, the highest among all renewable energy sources, due to the relatively faster reductions expected for solar investment costs.

4.1.3. Climate change mitigation and adaptation scenarios

When the impacts of global CC are integrated on top of the CCM and electrification assumptions, the results depict the optimal choice for shifting the Java-Bali power system to a low-carbon pathway while also responding to the effects of CC. The results showed that extra capacities are necessary to serve climate-induced electricity demand surges and to cope with decreased power-generating capacities (Fig. 4). Under CCMA RCP4.5, the extra capacities of solar PV, coal, wind, and biomass relative to CCM are 7.5%, 5.4%, 19.9%, and 0.7%, respectively. In contrast, the natural gas capacity reduces by 2.2% compared with CCM. Naturally, the capacities of solar PV, coal, wind, and biomass increase to balance the growth in energy demand and to compensate for CC-induced declines in the power-generating capacities.

³ Other parameters (e.g., wind speed, solar radiation) were assumed to remain unchanged.



Fig. 2. Capacity mix over the study period for the reference scenario (a) and the electricity mix in 2050 (b).



Solar Wind Biomass Geothermal Hydro Natural Gas Coal

Fig. 3. The Java-Bali electricity mixes for 2017, 2025, and 2050, under the climate change mitigation scenario (CCM).



Fig. 4. Extra capacity under CCMA RCP4.5 (a) and CCMA RCP8.5 (b) conditions, using the CCM scenario as a benchmark.

Under CCMA RCP8.5, the extra installed capacity is even higher than observed under CCMA RCP4.5, indicating that the higher temperature scenario intensifies the impacts on both electricity demand and power supply (e.g., coal, natural gas, and solar PV). The extra capacity under CCMA RCP8.5 is expected to be shared almost evenly among solar, coal, and wind (Fig. 4.b). The increased penetration of wind power indicates a trade-off between CC mitigation and adaptation. First, wind resources potentially remain available to facilitate meeting NRE targets, whereas the extra capacity of coal is constrained by the NRE target. Second, because our simulation excludes the impacts of CC on wind power, under the higher temperature scenario (CCMA RCP8.5), wind power becomes more competitive, replacing a portion of the capacity provided by solar PV under other scenarios. However, the role of solar PV remains significant when anticipating climate-induced surges in demand, adding 5% more capacity than observed for the CCM scenario.

The extra installed capacity necessary to anticipate CC adds 299 TWh and 388 TWh of electrical energy, respectively, under the CCMA RCP4.5 and CCMA RCP8.5 scenarios between 2020 and 2050. Under the CCMA RCP4.5 scenario, the additional energy is primarily provided



Fig. 5. Extra electricity production under CCMA RCP4.5 (a) and CCMA RCP8.5 (b) scenarios, using the CCM scenario as a benchmark.



Fig. 6. CO2 emissions (a) and total production costs (b) under reference, mitigation, and adaption scenarios. The CCM scenario is used as a benchmark.

by coal and solar PV (Fig. 5.a). Although solar PV adds an extra capacity under the CCMA RCP8.5 scenario (Fig. 4.b), it produces 6% less energy than under the CCM scenario, likely due to reductions in its efficiency, which is compensated by a significant increase in wind power production (Fig. 5.b).

4.2. CO₂ emissions and total costs

4.2.1. CO2 emissions

Indonesia's NDC outlines the country's target of reducing 29% of its emission from business as usual (BAU) by 2030, and the energy sector is targeted to contribute 19% of the targeted CO_2 reductions. These simulation results revealed that under the REF scenario, the Java-Bali power system will emit 215 million tonnes of CO_2 in 2030 (Fig. 6.a). Meanwhile, under the CCM scenario, the CO_2 emissions are expected to decrease to 188 million tonnes, which represents a 13% reduction compared with the REF scenario. Therefore, additional efforts beyond the implementation of the NRE policy target are necessary to achieve the power sector's NDC target by 2030. By 2050, however, the CO_2 emissions are expected to drop further, to 34% lower than the REF scenario. Therefore, when the NRE target for 2050 has been fulfilled, Indonesia will be able to reduce its emissions beyond its current NDC targets.

When compared with the CCM scenario, the CO_2 emissions are expected to increase for both CCMA scenarios, with 20 and 25 million tonnes CO_2 expected for CCMA RCP4.5 and RCP8.5, respectively (Fig. 6.a). The additional CO_2 emissions are attributable to the expected increase in electricity production from coal, which is necessary to compensate for the climate-induced surge in electricity demand. Moreover, CFPP efficiencies drop under CC conditions, requiring additional coal consumption per unit energy produced. However, even

with this increase, the CO_2 emissions are still expected to decrease by 30% compared with the REF scenario in 2050.

4.2.2. Total costs

Total costs were calculated based on the net present value of the total costs over the entire study period, which included capital costs, operation, and maintenance costs, and fuel costs. Under the REF scenario, the total costs of electricity production over the study period, from 2018 to 2050, are expected to reach 70.4 billion USD, which is equal to 0.1% of the cumulative national GDP during the study period⁴.

Under the CCM scenario, the total costs of expanding the Java-Bali power system from 2018 to 2050 are expected to reach 75.1 billion USD. Therefore, shifting from the current regime to the low-carbon pathway is expected to add 4.7 billion USD to cost projections for the REF scenario, resulting in 6.8% higher costs compared with the REF scenario. Furthermore, when CC impacts are considered, the costs grow by 10.7% and 11.4%, under the CCMA RCP4.5 and CCMA RCP8.5 scenarios, respectively, relative to the REF scenario. These results indicated that shifting to the low-carbon pathway while coping with CC impacts is expected to increase sector costs relative to the no mitigation-adaptation scenario (REF) (Fig. 6.b). Thus, the CC adaptation costs for the Java-Bali power system are 2.7 and 3.2 billion USD for CCMA RCP4.5 and RCP8.5, respectively, above the CC mitigation costs (compare CCMA RCP4.5 and CCMA RCP8.5 with CCM in Fig. 6.b). These costs are high, constituting more than half of the budget allocated by the Indonesian Government for CC mitigation and adaptation for all sectors in 2017. These additional costs are due to the extra installed capacity for coal, solar PV, and wind power that are necessary to cope with future CC.

⁴ Assuming annual GDP growth of 4.5%.

5. Discussion

The integration of CC mitigation and adaptation objectives into simulations of the Java-Bali power system expansion revealed the following findings. First, the Java-Bali technology composition will be significantly altered, driven by the constraints set in LEAP, including the minimum renewable energy share and adverse CC impacts. Second, the Java-Bali installed capacity will need to increase to meet the climate-induced increases in electricity demand and to compensate for losses in the reliability of Java-Bali power generation, due to new climate conditions. Finally, although CO_2 emissions will drop with mitigation efforts, the total costs of electricity production are expected to increase by 11.4% in 2050, due to mitigation and adaptation measures.

The NRE targets are expected to force the power system to shift to a low-carbon pathway, resulting in a 34% reduction in CO₂ emissions by 2050. The primary driver of the energy transition in Indonesia is expected to be solar PV, due to its vast remaining exploitable potential, whereas the exploitable potential of other renewables, such as hydro and geothermal, are limited. Likewise, solar PV is also expected to play a significant role in compensating for reduced capacity and energy generation caused by adverse CC impacts, especially under the CCMA RCP4.5 scenario. However, under the higher-temperature change scenario (CCMA RCP8.5), solar PV is expected to have reduced reliability, which will likely be compensated by wind power. However, the rapid deployment of solar PV will also produce waste, generated by retired solar PV systems, associated with the potential leaching of hazardous substances (e.g., lead and cadmium). PV recycling has been suggested as a potential method for preventing these environmental impacts, increasing CO₂ savings [73], and decreasing the energy payback time [74].

Meanwhile, coal is expected to play significant roles in satisfying the climate-induced increases in electricity demand and compensating for declines in power-generating capacities under both CCMA scenarios, resulting in increased CO_2 emissions compared with the CCM scenario. These findings clearly show the interplay between CC mitigation and adaptation objectives and suggest that CC impacts might hinder the CC mitigation efforts of the power system, providing the data-supported quantitative estimate for the hypotheses presented in the literature [75,76,3].

Because CC impacts are unavoidable due to past CO_2 emissions and delayed mitigation actions, these findings suggest several courses of action for policymakers. First, the recognition of the power sector's vulnerability to CC should be a key policy priority, allowing the sector to prepare adaptation action plans and to further integrate these plans into national adaptation strategies. Second, sufficient budget allocations are necessary to facilitate the power sector's adaptations to CC. Since the power sector's adaptations for CC are likely to hinder CO_2 mitigation efforts, a nexus between mitigation and adaptation policies should be exploited in the national budget allocation. Finally, more considerable efforts are necessary to conduct detailed investigations of the CC impacts on individual power plants, including full analyses of the costs and benefits of their adaptations, which may include local investments to prevent reductions in capacity factor and efficiency at individual vulnerable power plants.

6. Conclusions

The primary goal of the present study was to integrate both CC mitigation and adaptation objectives into simulations of a long-term power system expansion. We use Indonesia as an example of a developing country that is currently struggling to meet its electrification needs. The primary power system, Java-Bali, was the focus of this study. Four scenarios for the Java-Bali power system expansion from 2018 through to 2050 were analyzed, including a reference scenario, a CC mitigation scenario, and two scenarios that integrate CC mitigation

and adaptation. For all CC mitigation scenarios, the renewable energy targets declared by the Indonesian government, which aim to increase the share of new and renewable energy to 23%, in 2025, and to 31%, in 2050, were used.

The impacts of CC on power supply and demand were quantified using various methods. The WEAP model was used to quantify CC impacts on hydropower availability, and empirical data from fieldwork were combined with reports from the literature to assess the CC impacts on TPPs and electricity demand. The estimated impacts included changes in HEPP capacity factors, reductions in CFPP efficiencies and capacity factors, reductions in NGPP capacity factors, reductions in solar PV generating capacities, and increases in electricity demands. Furthermore, these quantified impacts were integrated into the Java-Bali power system expansion simulation using LEAP. The simulation results were discussed in terms of installed capacity, electricity mix, costs, and CO_2 emissions.

This study provided insight into the economically optimal options for electricity generation by the Java-Bali power system under various CC mitigation-adaptation scenarios. CC-associated impacts on both supply and demand were considered. Although this study did not address the CC impacts for all types of power plants, the results indicated significant deviations from the REF scenario for installed capacity, electricity generation, and the total costs of electricity production, which were attributable to CC mitigation and adaptation strategies. Although the impacts of extreme weather events were excluded, interestingly, the costs of power system adaptations to gradual changes in climate variables, such as temperature and precipitation, were found to be comparable to the costs of CC mitigation.

The methodological approach utilized by this study could be replicated by other attempts to analyze the interplay between CC mitigation and adaptation in the power sector. The present study, however, has several limitations, primarily with regards to the assessment of CC impacts on power generation. First, the projected CC impacts on the Indonesian power sector are preliminary, driven by assessments found in the literature, rather than case-specific, downscaled, CC scenarios. However, this secondary data set provided the basis for an initial evaluation of the CC impacts on the power sector. Furthermore, this study generalized the impacts of CC on CFPPs and HEPPs based on our fieldwork data from several power plants. Future work should focus on collecting detailed data for individual power plants to assess the sitespecific costs of CC and possible adaptation solutions. Second, our modeling approach did not include the impacts of extreme weather events, which may intensify under future CC conditions, causing physical infrastructure damage, energy losses, and indirect effects, such as business interruption. Exploring these possibilities would be a fruitful area for further work. Third, due to data limitations and the scope of LEAP, CC impacts on transmission and distribution of electricity were excluded from this assessment while they appear to be significant [44]. Further research should quantify the CC impacts on all components of the power sector: generation, transmission, and distribution. Finally, this analysis relied on global CC projections, which are highly uncertain. Therefore, the modeling of CC impacts on electricity supply and demand performed in this study are inherently uncertain, based on the chosen CC scenarios.

Despite these limitations, this study provided a methodological framework for performing an integrated analysis of CC mitigation and adaptation strategies for the power sector, allowing the interplay between CC mitigation and adaptation to be assessed, which is crucial for developing optimal mitigation and adaptation strategies. This framework can be extended to account for other CC impacts, such as extreme weather events, the impacts of gradual changes in climate variables on transmission and distribution networks, and the impacts on other types of electric power technologies, such as wind power, geothermal, nuclear, and storage technologies.

CRediT authorship contribution statement

Kamia Handayani: Conceptualization, Methodology, Formal analysis, Writing - original draft. Tatiana Filatova: Conceptualization, Supervision, Writing - review & editing. Yoram Krozer: Conceptualization, Supervision. Pinto Anugrah: Formal analysis.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Appendix A. Overview of the LEAP energy system model

Applied Energy 262 (2020) 114485

Acknowledgment

We gratefully acknowledge the financial support provided by the Indonesian Endowment Fund for Education (LPDP) under the Grant number PRJ-2570/LPDP/2015. Furthermore, we would like to express our gratitude to PT Perusahaan Listrik Negara, PT Pembangkitan Jawa Bali, and PT Indonesia Power, the World Bank Climate Knowledge Portal, and the Asia-Pacific Data Research Center for providing data for our modeling assumptions. We also extend our appreciation to the Stockholm Environmental Institute for their support regarding the LEAP software.

The essential features of LEAP for addressing the primary goal of this article includes its support for alternative scenario projections, its least-cost optimization modeling of power system expansion, and the calculation of CO_2 emissions.

LEAP consists of three modules: I. Demand, II. Transformation and III. Resources modules (the black boxes of Fig. A1). The power system simulation called the electricity generation module in LEAP belongs to the transformation module. The electricity generation module simulates electricity supply to satisfy the given demand, based on various input parameters. Accordingly, the resources module calculates the required fuel to generate electricity simulated in the transformation module. The model outputs, which are of most interest given the goal of this article, is the added capacity and electricity generation for each technology, CO₂ emissions, and costs.

The simulation of electricity generation consists of three steps. First, LEAP calculates the capacity expansion required to satisfy the demand and the capacity reserve of the power system. The outputs of this calculation are the capacity added each year and the composition of technology (capacity mix). Second, LEAP dispatches electricity from each process in accordance with the annual demand and the load curve. The output of the second step is the annual electricity production from each process. Three, the resource module calculates the primary energy required to generate electricity based on the fuel efficiency of each technology. Additionally, LEAP calculates the power system's total costs based on the costs' input data. Moreover, LEAP includes technology and environmental database that allows the calculation of CO2 emissions from the electricity production based on the IPCC Tier 1 emission factor.

In LEAP, the optimal solution is defined as the power system with the lowest total net present value of the total costs over the entire period of calculation (from the base year to the end year). Thus, the optimization setting works through integration with the Open Source Energy Modelling System (OSeMOSYS). LEAP automatically writes the data files required by OSeMOSYS, making use of the same data that were input into LEAP. The results of the optimization are also read back into LEAP so that all relevant results can be viewed in LEAP. In turn, OSeMOSYS depends on a solver software tool for developing decision optimization models. Due to the complexity of the simulations performed in this article, a more powerful solver, namely the CPLEX optimizer, which is a software toolkit developed by IBM, was used instead of the LEAP built-in GNU Linear Programming Kit.

One limitation in LEAP is that the model does not provide for the expansion of transmission and distribution lines. Hence, the power system expansion simulation in this article neglected constraints in the T&D networks, assuming that electricity can be transmitted at any time to any load station. Further modeling works will have to cover transmission capacity and spatial analysis of each power plant and substation.



Fig. A1. LEAP-WEAP model for Integrated climate change mitigation-adaptation analysis.

Appendix B. Projection of temperature and precipitation changes

(See. Tables B1-B2)

Table B1

Historical and projected changes in temperature.

Month	Temperature (°C) 1986–2005	Projected changed					
		2020–2039		2040–2059			
		RCP 4.5	RCP 8.5	RCP 4.5	RCP 8.5		
January	25.88	0.69	0.76	1.03	1.37		
February	25.94	0.73	0.82	1.09	1.45		
March	26.22	0.72	0.81	1.06	1.40		
April	26.43	0.75	0.83	1.05	1.42		
May	26.41	0.71	0.82	1.07	1.43		
June	26.05	0.72	0.81	1.07	1.45		
July	25.73	0.73	0.84	1.06	1.42		
August	25.74	0.71	0.82	1.07	1.41		
September	25.92	0.77	0.82	1.10	1.43		
October	26.20	0.77	0.83	1.09	1.43		
November	26.21	0.74	0.84	1.09	1.41		
December	25.95	0.69	0.79	1.02	1.39		

Table B2

-

Historical and projected changes in precipitation.

Month	Precipitation (mm) 1986–2005	Projected changed					
		2020–2039		2040–2059			
		RCP 4.5	RCP 8.5	RCP 4.5	RCP 8.5		
January	336.73	18.08	9.42	29.90	20.87		
February	266.76	-3.62	4.48	0.52	14.48		
March	343.74	0.83	15.38	15.36	15.56		
April	327.12	-5.28	4.32	10.43	13.86		
May	240.61	-6.10	-1.65	-0.06	1.48		
June	115.71	-3.43	-7.23	-0.25	-3.47		
July	109.90	-8.02	-7.95	-10.32	-17.47		
August	101.13	-1.69	-0.22	-6.22	- 3.98		
September	137.69	-4.24	-2.18	-2.16	-6.87		
October	287.00	-8.80	-9.24	-7.22	-10.95		
November	398.00	1.99	8.53	13.27	4.36		
December	340.32	-1.83	-8.89	15.28	-5.09		

Appendix C. Assumptions for the LEAP model

(See. Table C1)

Table C1

Renewable energy potential and current practice in Indonesia.

07 1	-				
Renewable	Potential in Gigaw	vatt	Renewable deployment by 2018, total Indonesia		Sources
	Total Indonesia	Java- Bali islands	Installed capacity (Gigawatt)	Renewable utilization (%)	
Hydro	75	4.2	5.4	7.2%	DEN [55], ESDM [77]
Hydro pumped storage	4.3	3.9	0	0.0%	PLN [78], ESDM [77]
Mini hydro	19.4	2.9	0.4	2.1%	DEN [55], ESDM [77]
Geothermal	17.5 ^a	6.8 ^a	1.9	11%	DEN [55], ESDM [77]
Biomass	30	7.4	1.8	6.0%	DEN [55], ESDM [77]
Solar	5374 ^b	2747 ^b	0.06	0.0%	Kunaifi and Reinders [79], ESDM [77]
Wind	60.6	24.1	0.14	0.2%	DEN [55], ESDM [77]

^a excluding the speculative and hypothetical potential.

^b in Gigawatt peak.

Appendix D. Assumptions for projecting climate change impacts on the electricity supply





Fig. D1. Schematic view of water demand and supply of the Citarum WEAP model. Source: own simulations.



Fig. D2. The average annual capacity factor of HEPPs under CCM and CCMA scenarios: (a) Saguling HEPP; (b) Cirata HEPP. Source: WEAP estimates.

Table D1

Input data for the Citarum WEAP model.

Input Data	Value	Sources
Catchment		
Мар	Fig. D1	KLHK [80], BIG [81]
Area	Saguling: 222 thousand ha,	Own calculation based on
	Cirata: 416 thousand ha	geospatial analysis
Precipitation (mm)	Monthly, 1948–2017	Sheffield et al. [82],
		PLTA Cirata [83], and PLTA
		Saguling [84]
Temperature (°C)	Mean monthly air temp,	Sheffield et al. [82]
	1948–2010	
Wind speed (m/s)	Monthly, 1948–2010	Sheffield et al. [82]
Total Storage Capacity	Cirata: 2,165 million m ³ ,	PLTA Cirata [83],
	Saguling: 875 million m ³	PLTA Saguling [84]
Max Turbine Flow	Cirata: 1,080 m ³ /s,	PLTA Cirata [83],
	Saguling: 219.2 m ³ /s	PLTA Saguling [84]
Tailwater Elevation	Cirata: 84 m,	Kasiro et al. [85]
	Saguling: 252 m	

References

- [1] UNFCCC. Aggregate effect of the intended nationally determined contributions: an update synthesis report by the secretariat; 2016.
- [2] Schaeffer R, Szklo AS, Pereira de Lucena AF, Moreira Cesar Borba BS, Pupo Nogueira LP, Fleming FP, et al. Energy sector vulnerability to climate change: a review. Energy 2012;38:1–12. https://doi.org/10.1016/j.energy.2011.11.056.
- [3] Cronin J, Anandarajah G, Dessens O. Climate change impacts on the energy system: a review of trends and gaps. Clim Change 2018;151:79–93. https://doi.org/10. 1007/s10584-018-2265-4.
- [4] Grande-Acosta G, Islas-Samperio J. Towards a low-carbon electric power system in Mexico. Energy Sustain Dev 2017;37:99–109. https://doi.org/10.1016/J.ESD.2017. 02.001.
- [5] Handayani K, Krozer Y, Filatova T. Trade-offs between electrification and climate change mitigation: an analysis of the Java-Bali power system in Indonesia. Appl Energy 2017;208. https://doi.org/10.1016/j.apenergy.2017.09.048.
 [6] Dalla Longa F, van der Zwaan B. Do Kenya's climate change mitigation ambitions
- [6] Dalla Longa F, van der Zwaan B. Do Kenya's climate change mitigation ambitions necessitate large-scale renewable energy deployment and dedicated low-carbon energy policy? Renew Energy 2017;113:1559–68. https://doi.org/10.1016/J. RENENE.2017.06.026.
- [7] Haiges R, Wang YD, Ghoshray A, Roskilly AP. Optimization of Malaysia's power generation mix to meet the electricity demand by 2050. Energy Procedia 2017;142:2844–51. https://doi.org/10.1016/J.EGYPRO.2017.12.431.
- [8] Guo Z, Cheng R, Xu Z, Liu P, Wang Z, Li Z, et al. A multi-region load dispatch model for the long-term optimum planning of China's electricity sector. Appl Energy 2017;185:556–72. https://doi.org/10.1016/j.apenergy.2016.10.132.
- [9] Audinet P, Amado J-C, Rabb B. Climate Risk Management Approaches in the Electricity Sector: Lessons from Early Adapters; 2014:17–64. http://doi.org/10. 1007/978-1-4614-9221-4 2.
- [10] Clarke L, Sokka L, Jiang K, Akimoto K, Babiker M, Blanford G, et al. Assessing Transformation Pathways Coordinating Lead Authors: Chapter Science Assistant. Clim Chang 2014 Mitig Clim Chang Contrib Work Gr III to Fifth Assess Rep Intergov Panel Clim Chang; 2014:98.
- [11] Field CB, Barros VR, Mach KJ, Mastrandrea MD, van Aalst M, Adger WN, et al. Technical summary. Clim Chang 2014 Impacts. Adapt Vulnerability Part A Glob Sect Asp Contrib Work Gr II to Fifth Assess Rep Intergov Panel Clim Chang 2014;2014:35–94.
- [12] Tobin I, Greuell W, Jerez S, Ludwig F, Vautard R, Van Vliet MTH, et al. Vulnerabilities and resilience of European power generation to 1.5 °C, 2 °C and 3 °C warming. Environ Res Lett 2018;13.. https://doi.org/10.1088/1748-9326/aab211.
- [13] Kern JD, Characklis GW. Evaluating the Financial Vulnerability of a Major Electric Utility in the Southeastern U.S. to Drought under Climate Change and an Evolving Generation Mix. Environ Sci Technol 2017;51:8815–23. https://doi.org/10.1021/ acs.est.6b05460.
- [14] Cook MA, King CW, Davidson FT, Webber ME. Assessing the impacts of droughts and heat waves at thermoelectric power plants in the United States using integrated regression, thermodynamic, and climate models. Energy Rep 2015;1:193–203. https://doi.org/10.1016/j.egyr.2015.10.002.
- [15] van Vliet MTH, Yearsley JR, Ludwig F, Vogele S, Lettenmaier DP, Kabat P. Vulnerability of US and European electricity supply to climate change. Nat Clim Chang 2012;2:676–81.
- [16] Zheng X, Wang C, Cai W, Kummu M, Varis O. The vulnerability of thermoelectric power generation to water scarcity in China: current status and future scenarios for power planning and climate change. Appl Energy 2016;171:444–55. https://doi. org/10.1016/j.apenergy.2016.03.040.
- [17] Jerez S, Tobin I, Vautard R, Montávez JP, López-Romero JM, Thais F, et al. The impact of climate change on photovoltaic power generation in Europe. Nat Commun 2015;6:10014. https://doi.org/10.1038/ncomms10014.
- [18] Pereira de Lucena AF, Szklo AS, Schaeffer R, Dutra RM. The vulnerability of wind

power to climate change in Brazil. Renew. Energy 2010;35:904–12. https://doi. org/10.1016/J.RENENE.2009.10.022.

- [19] Pašičko R, Branković Č, Šimić Z. Assessment of climate change impacts on energy generation from renewable sources in Croatia. Renew Energy 2012;46:224–31. https://doi.org/10.1016/J.RENENE.2012.03.029.
- [20] Lise W, van der Laan J. Investment needs for climate change adaptation measures of electricity power plants in the EU. Energy Sustain Dev 2015;28:10–20. https://doi. org/10.1016/J.ESD.2015.06.003.
- [21] Anugrah P, Setiawan AA, Budiarto R, Sihana. Evaluating Micro Hydro Power Generation System under Climate Change Scenario in Bayang Catchment, Kabupaten Pesisir Selatan, West Sumatra. Energy Procedia 2015;65:257–63. https://doi.org/10.1016/j.egypro.2015.01.043.
- [22] Guerra OJ, Tejada DA, Reklaitis GV. Climate change impacts and adaptation strategies for a hydro-dominated power system via stochastic optimization. Appl Energy 2019;233–234:584–98. https://doi.org/10.1016/j.apenergy.2018.10.045.
- [23] Boehlert B, Strzepek KM, Gebretsadik Y, Swanson R, McCluskey A, Neumann JE, et al. Climate change impacts and greenhouse gas mitigation effects on U.S. hydropower generation. Appl Energy 2016;183:1511–9. https://doi.org/10.1016/j. apenergy.2016.09.054.
- [24] de Lucena AFP, Schaeffer R, Szklo AS. Least-cost adaptation options for global climate change impacts on the Brazilian electric power system. Glob Environ Chang 2010;20:342–50. https://doi.org/10.1016/j.gloenvcha.2010.01.004.
- [25] Teotónio C, Fortes P, Roebeling P, Rodriguez M, Robaina-Alves M. Assessing the impacts of climate change on hydropower generation and the power sector in Portugal: a partial equilibrium approach. Renew Sustain Energy Rev 2017;74:788–99. https://doi.org/10.1016/j.rser.2017.03.002.
- [26] Arango-Aramburo S, Ríos-Ocampo JP, Álvarez-Espinosa AC, Kober T, Turner SWD, Daenzer K, et al. Climate impacts on hydropower in Colombia: A multi-model assessment of power sector adaptation pathways. Energy Policy 2019;128:179–88. https://doi.org/10.1016/j.enpol.2018.12.057.
- [27] Spalding-Fecher R, Joyce B, Winkler H. Climate change and hydropower in the Southern African Power Pool and Zambezi River Basin: System-wide impacts and policy implications. Energy Policy 2017;103:84–97. https://doi.org/10.1016/J. ENPOL.2016.12.009.
- [28] Baxter LW, Calandri K. Global warming and electricity demand: a study of California. Energy Policy 1992;20:233–44. https://doi.org/10.1016/0301-4215(92)90081-C.
- [29] Sailor DJ, Pavlova AA. Air conditioning market saturation and long-term response of residential cooling energy demand to climate change. Energy 2003;28:941–51. https://doi.org/10.1016/S0360-5442(03)00033-1.
- [30] Mansur ET, Mendelsohn R, Morrison W. Climate change adaptation: A study of fuel choice and consumption in the US energy sector. J Environ Econ Manage 2008;55:175–93. https://doi.org/10.1016/J.JEEM.2007.10.001.
- [31] Cartalis C, Synodinou A, Proedrou M, Tsangrassoulis A, Santamouris M. Modifications in energy demand in urban areas as a result of climate changes: an assessment for the southeast Mediterranean region. Energy Convers Manage 2001;42:1647–56. https://doi.org/10.1016/S0196-8904(00)00156-4.
- [32] Lam TNT, Wan KKW, Wong SL, Lam JC. Impact of climate change on commercial sector air conditioning energy consumption in subtropical Hong Kong. Appl Energy 2010;87:2321–7. https://doi.org/10.1016/j.apenergy.2009.11.003.
- [33] Totschnig G, Hirner R, Müller A, Kranzl L, Hummel M, Nachtnebel HP, et al. Climate change impact and resilience in the electricity sector: the example of Austria and Germany. Energy Policy 2017;103:238–48. https://doi.org/10.1016/j. enpol.2017.01.019.
- [34] Dowling P. The impact of climate change on the European energy system. Energy Policy 2013;60:406–17. https://doi.org/10.1016/j.enpol.2013.05.093.
- [35] Seljom P, Rosenberg E, Fidje A, Haugen JE, Meir M, Rekstad J, et al. Modelling the effects of climate change on the energy system—A case study of Norway. Energy Policy 2011;39:7310–21. https://doi.org/10.1016/J.ENPOL.2011.08.054.
- [36] Mima S, Criqui P. The costs of climate change for the European energy system, an

assessment with the POLES model. Environ Model Assess 2015;20:303–19. https://doi.org/10.1007/s10666-015-9449-3.

- [37] Li S, Coit DW, Selcuklu S, Felder FA. Electric Power Generation Expansion Planning: Robust Optimization Considering Climate Change. In: Proc 2014 Ind Syst Eng Res Conf 2014:1–10. http://doi.org/10.1021/acschembio.5b00762.
- [38] PLN. Statistik PLN 2016 2017. https://www.pln.co.id/stakeholder/laporan-statistik [accessed April 18, 2018].
- [39] BPS. Sensus Penduduk 2010. Biro Pusat Statistik Indonesia; 2010.
- [40] Handayani K, Krozer Y, Filatova T. From fossil fuels to renewables: An analysis of long-term scenarios considering technological learning. Energy Policy 2019:134–46. https://doi.org/10.1016/j.enpol.2018.11.045.
- [41] Heaps CG. Long-range Energy Alternatives Planning (LEAP) system. [Software version 2017.0.5] 2017. http://www.energycommunity.org.
- [42] Khan SI, Islam A, Khan AH. Energy Forecasting of Bangladesh in Gas Sector Using LEAP Software. Glob J Res Eng 2011;11:6.
- [43] Connolly D, Lund H, Mathiesen BV, Leahy M. A review of computer tools for analysing the integration of renewable energy into various energy systems. Appl Energy 2010;87:1059–82. https://doi.org/10.1016/j.apenergy.2009.09.026.
- [44] Handayani K, Filatova T, Krozer Y. The vulnerability of the power sector to climate variability and change: evidence from Indonesia. Energies 2019;12:3640. https:// doi.org/10.3390/en12193640.
- [45] Zhang Y, Gu A, Lu H, Wang W. Hydropower generation vulnerability in the Yangtze River in China under climate change scenarios: analysis based on the WEAP model. Sustain 2017;9. http://doi.org/10.3390/su9112085.
- [46] The World Bank Group. Climate Change Knowledge Portal. Country: Indonesia 2019. https://climateknowledgeportal.worldbank.org/country/indonesia (accessed April 18, 2019).
- [47] Harris I, Jones PD, Osborn TJ, Lister DH. Updated high-resolution grids of monthly climatic observations - the CRU TS3.10 Dataset. Int J Climatol 2014;34:623–42. https://doi.org/10.1002/joc.3711.
- [48] International Pacific Research Center. CIMP5 dataset 2018. http://apdrc.soest. hawaii.edu/las/v6/dataset?catitem = 3228 [accessed May 13, 2019].
- [49] PLN. Electricity Supply Bussiness Plan (RUPTL) 2018-2027 2018. http://www.pln. co.id/stakeholder/ruptl [accessed June 18, 2018].
- [50] OECD/IEA. World Energy Outlook World Energy Model 2017; 2017. http://www. worldenergyoutlook.org/weomodel/.
- [51] DEN. Outlook Energi Indonesia 2014; 2014.
- [52] KESDM. Draft Rencana Umum Ketenagalistrikan Nasional 2015-2034; 2015.
- [53] JICA. Preparatory Survey for Indramayu Coal-fired Power Plant Project in Indonesia; 2010.
- [54] PLN. RUPTL cost assumptions; 2017.
- [55] DEN. Indonesia Energy Outlook 2016. National Energy Council; 2016.
- [56] PLN. PLN Statistics 2015. Jakarta: Corporate Secretary PT PLN (Persero); 2016.
- [57] ACE. Levelised cost of electricity of selected renewable technologies in the ASEAN member states. Jakarta: ASEAN Center for Energy; 2016.
- [58] Rothwell G, Rust J. On the optimal lifetime of nuclear power plants. J Bus Econ Stat 1997;15:195–208. https://doi.org/10.2307/1392305.
- [59] IEA, NEA. Projected Costs of Generating Electricity 2015 Edition. Paris: International Energy Agency Nuclear Energy Agency; 2015.
- [60] IRENA. Renewable energy technologis: Cost analysis series. vol. 1. Abu Dhabi; 2012
- [61] BMKG. Temperature data 2014 2018. http://dataonline.bmkg.go.id/ [accessed February 2, 2018].
- [62] Spalding-fecher DR. Impact of climate change and irrigation development on hydropower supply in the Zambezi River Basin , and implications for power sector

development in the Southern African Power Pool; 2018.

- [63] Allen RG, Pereira LS, Raes D, Smith M. Crop evapotranspiration: Guidelines for computing crop requirements. Irrig Drain Pap No 56, FAO 1998:300. http://doi. org/10.1016/j.eja.2010.12.001.
- [64] Sieber J. Water Evaluation And Planning (WEAP) System, Software version: 2019.0, Stockholm Environment Institute, Somerville, MA, USA 2019. https://www. weap21.org [accessed February 21, 2019].
- [65] Yates D, Sieber J, Purkey D, Huber-Lee A. WEAP21 A demand-, priority-, and preference-driven water planning model. Part 1: Model characteristics. Water Int 2005;30:487–500. https://doi.org/10.1080/02508060508691893.
- [66] Haldkar V, Kumar Sharma A, Ranjan RK, Bajpai VK. An energy analysis of condenser. Int J Therm Technol 2013;3:120–5.
- [67] Henry CL, Pratson LF. Effects of environmental temperature change on the efficiency of coal- and natural gas-fired power plants. Environ Sci Technol 2016;50:9764–72. https://doi.org/10.1021/acs.est.6b01503.
- [68] Drbal L, Westra K, Boston P. Power Plant Engineering. Springer, Boston, MA; 1996. http://doi.org/10.1007/978-1-4613-0427-2.
- [69] Kehlhofer R, Rukes B, Hannemann F, Stirnimann F. Combined-Cycle Gas & Steam Turbine Power Plants. PennWell Corp 2009.
- [70] Asian Development Bank. Adaptation to Climate Change The Case of a Combined Cycle Power Plant; 2012.
- [71] Simioni T, Schaeffer R. Georeferenced operating-efficiency solar potential maps with local weather conditions – an application to Brazil. Sol Energy 2019;184:345–55. https://doi.org/10.1016/J.SOLENER.2019.04.006.
- [72] Dubey S, Sarvaiya JN, Seshadri B. Temperature dependent photovoltaic (PV) efficiency and its effect on PV production in the world – a review. Energy Procedia 2013;33:311–21. https://doi.org/10.1016/J.EGYPRO.2013.05.072.
- [73] D'Adamo I, Miliacca M, Rosa P. Economic feasibility for recycling of waste crystalline silicon photovoltaic modules. Int J Photoenergy 2017;2017. https://doi.org/ 10.1155/2017/4184676.
- [74] Goe M, Gaustad G. Strengthening the case for recycling photovoltaics: an energy payback analysis. Appl Energy 2014;120:41–8. https://doi.org/10.1016/j. apenergy.2014.01.036.
- [75] Fisher-Vanden K, Sue Wing I, Lanzi E, Popp D. Modeling climate change feedbacks and adaptation responses: recent approaches and shortcomings. Clim Change 2013;117:481–95. https://doi.org/10.1007/s10584-012-0644-9.
- [76] Turner SWD, Hejazi M, Kim SH, Clarke L, Edmonds J. Climate impacts on hydropower and consequences for global electricity supply investment needs. Energy 2017;141. https://doi.org/10.1016/j.energy.2017.11.089.
- [77] ESDM. Statistik Ketenagalistrikan 2018. Jakarta, Indonesia: Directorate General of Electricity, Ministry of Energy and Mineral Resources; 2019.
- [78] PLN. Electricity Supply Business Plan 2017-2026 (RUPTL); 2017.
- [79] Kunaifi K, Reinders A. Opportunities for Photovoltaic Solar Energy Systems in Indonesia. WREC-15 2016.
- [80] KLHK. Peta DAS Prioritas Citarum. Bogor: KLHK; 2011.
- [81] BIG. Ina-Geoportal 2017. http://tanahair.indonesia.go.id/portal-web [accessed December 6, 2018].
- [82] Sheffield J, Goteti G, Wood EF. Development of a 50-Year High-Resolution Global Dataset of Meteorological Forcings for Land Surface Modeling. J Clim 2006;19:3088–111. https://doi.org/10.1175/JCLI3790.1.
- [83] PLTA Cirata. Fieldwork data; 2018.
- [84] PLTA Saguling. Fieldwork data; 2018.
- [85] Ibnu Kasiro, Isdiana, D. Pangluar, CL Nugroho, Attan Muchtar, Hari Martadi RS. Bendungan Besar di Indonesia. Pusat Penelitian dan Pengembangan Pengairan, Dept. PU, Dirjen Pengairan; 1995.